Author’s response to reviews

Title: Association of caries experience and dental plaque with sociodemographic characteristics in elementary school-aged children: a cross-sectional study

Authors:
Saeed bashirian (Bashirian@umsha.ac.ir)
samane shirahmadi (shirahmadi.S@umsha.ac.ir; shirahmadi_s@yahoo.com)
Shabnam Seyedzadeh-Sabounchi (shsab1@gmail.com)
alireza soltanian (Soltanian@umsha.ac.ir)
akram karimi-shahanjarini (karimi.a@umsha.ac.ir)
Farshid vahdatinia (Fa.vahdatinia@gmail.com)

Version: 6 Date: 08 Dec 2017

Author’s response to reviews:

Title: Association of caries experience and dental plaque with sociodemographic characteristics in elementary school-aged children: a cross-sectional study

Authors:
Saeed Bashirian (s-bashirian@umsha.ac.ir)
Samaneh Shirahmadi (S.shirahmadi@umsha.ac.ir)
Shabnam Seyedzadeh Sabounchi, (shsab1@ gmail.com)
Ali Reza Soltanian (Soltanian@ umsha.ac.ir)
Akram Karimi-shahanjarini, (karimi.a@umsha.ac.ir)
Farshid Vahdatinia (Fa.vahdatinia@gmail.com)
Author's response to reviews: see over

1) Author Contributions

Please represent authors' names using their full initials, not their full name, in the Authors’ Contributions section. We note that two authors have the same initials (SS). To distinguish between them in the Authors’ Contributions section please designate them as SS1 and SS2, with SS1 corresponding to Samane Shirahmadi, chosen because this name is furthest up the author list.

Response: Thank you for your valuable comment. We have revised authors’ contributions section (page 18, lines 16-20).

2) Clean Manuscript

At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

Response: We have uploaded a clean version of the manuscript and figure without tracked changes or highlighting.